Title: A controlled trial of automated classification of negation from clinical notes

Authors:

Peter L Elkin (elkin.peter@mayo.edu)
Steven H Brown (Steven.Brown@med.va.gov)
Brent A Bauer (bauer.brent@mayo.edu)
Casey S Husser (husser.casey@mayo.edu)
Bill Carruth (bcarruth@jhmi.edu)
Larry R Bergstrom (bergstrom.larry@mayo.edu)
Dietlind L Wahner-Roedler (wahnerroedler.dietlind@mayo.edu)

Version: 8 Date: 29 April 2005

Author's response to reviews: see over
April 29, 2005

Iratxe Puebla, Asst. Editor
BioMed Central
Middlesex House
34-42 Cleveland Street
London W1T 4LB

MS: 1273360723448134
A Controlled Trial of Automated Classification of Negation from Clinical Notes

Dear Assistant Editor:

I have made the following format changes as requested:

**Competing interests**- I have removed the paragraph regarding to the availability of the software from the competing interest declaration and include it within the text of the manuscript, under the Methods and next to a reference to the software.

**Additional files**- I have removed the section referring to additional files from page11.

Should you have any questions please do not hesitate to contact me.

Sincerely,

Peter L. Elkin, MD
Professor of Medicine
Mayo Clinic College of Medicine